Author's response to reviews

Title: Barriers faced by Ugandan university students in seeking medical care and sexual health counselling: A cross-sectional study

Authors:

Andualem T Boltena (andy_tadesse@yahoo.com)
Farhad A Khan (farhad.khan@smi.se)
Benedict O Asamoah (benedict_oppo@med.lu.se)
Anette Agardh (anette.agardh@med.lu.se)

Version: 2 Date: 31 May 2012

Author's response to reviews: see over
To: BMC Public Health
Editorial Board

Dear Dr Audrey,

Please find attached the updated version of our manuscript, “Barriers faced by Ugandan university students in seeking medical care and sexual health counselling: A cross-sectional study”.

Below is the point by point description of the changes we have made in this version of the manuscript.

1. Different sections of the abstract have been separated out in the submission system.
2. The contributions of each author has been mentioned separately in the Author’s Contribution section.
3. The details of the ethical approval for this study have been provided in the method section.
4. This study was part of a larger study. We have attached the questionnaire.

Sincerely yours,

Andualem Tadesse Boltena (corresponding author)
Social Medicine and Global Health
Dept of Clinical Sciences, Malmö
Lund University, Sweden